\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage{comment}
\usepackage[colorlinks]{hyperref}


\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]


%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}

\def\udcs{517.545, 519.177} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{58J53, 05C76} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}



\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{  }} \\[2mm]
     {\LARGE \textsf{  }} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Èíñòèòóò ìàòåìàòèêè èì. Ñ.Ë.Ñîáîëåâà ÑÎ ÐÀÍ}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 20, No. 2, pp. 144--145 (2023) \hspace{\fill}{\rm\small UDK \ \ \udcs\hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
     	{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.}}\hphantom{aaaaaaaaaaaaaaaaa\;}{\rm\small MSC \ \ \mscs }}%\vbox
}


\begin{document}
\begin{thebibliography}{1} 
% Ññûëêè íà ëèòåðàòóðó ïðèâîäÿòñÿ òîëüêî íà àíãëèéñêîì ÿçûêå (äàæå åñëè ó èñòî÷íèêà íåò àíãëîÿçû÷íîé âåðñèè).

\bibitem{BardKang} D. Barden, H. Kang, 
\href{https://doi.org/10.1017/S0305004112000126}
{\it Isospectral surfaces of genus two and three}, 
Math. Proc. Camb. Phil. Soc., {\bf 153}:1 (2012), 99 --110. MR2943668

\bibitem{BrooksTse87} R. Brooks, R. Tse, 
\href{https://doi.org/10.1017/S0027763000002518}
{\it Isospectral surfaces of small genus}, 
Nagoya Math. J. {\bf 107} (1987), 13--24;  MR0909246
Corrigendum: R. Brooks, R. Tse, Nagoya Math. J. {\bf 117} (1990), 227.  MR1044942

\bibitem{Brooks88} R. Brooks, 
\href{https://doi.org/10.1080/00029890.1988.11972094}
{\it Constructing isospectral manifolds}, 
Amer. Math. Monthly, {\bf 95}:9 (1988), 823--839. MR0967343

\bibitem{Buser92} P. Buser, 
\href{https://doi.org/10.1007/978-0-8176-4992-0}
{\it Geometry and spectra of compact Riemann surfaces}, 
Progress in Mathematics, {\bf 106}, Birkh\"auser, Boston, MA, 1992. Zbl 1239.32001

\bibitem{Buser86} P. Buser, 
\href{http://eudml.org/doc/74711}
{\it Isospectral Riemann surfaces}, 
Ann. Inst. Fourier, {\bf 36} (1986), 167--192. MR0850750

\bibitem{ConwaySloan} J. H. Conway, N. J. A. Sloane, 
\href{https://doi.org/10.1155/S1073792892000102}
{\it Four-dimensional lattices with the same theta series}, 
Internat. Math. Res. Notices {\bf 4} (1992), 93--96. MR1159450

\bibitem{DamHaemers} E. R. van Dam, W. H. Haemers, 
\href{https://doi.org/10.1016/S0024-3795(03)00483-X}
{\it Which graphs are determined by their spectrum?}, 
Linear Algebra and its Applications, {\bf 373} (2003), 241--272. MR2022290

\bibitem{EarnNipp} A. G. Earnest,  G. Nipp, 
\href{https://mathreports.ca/article/on-the-theta-series-of-positive-quaternary-quadratic-forms/}
{\it On the theta series of positive quaternary quadratic forms}, C. R. Math. Rep. Acad. Sci. Canada, {\bf 13}:1 (1991), 33--38. MR1097501
 
\bibitem{Fiedler} 
M. Fiedler, \href{https://eudml.org/doc/12723}
{\it Algebraic connectivity of graphs}, 
Czech. Math. J., {\bf 23} (98) (1973) 298--305. MR0318007

\bibitem{Friedli} F. Friedli, 
\href{https://doi.org/10.4171/AIHPD/66}
{\it The bundle Laplacian on discrete tori}, 
Ann. Inst. Henri Poincar\'e, Comb. Phys. Interact. {\bf 6}:1 (2019), pp. 97--121. MR3911691

\bibitem{Godsil}
C.D. Godsil, D.A. Holton, B. McKay,
\href{https://doi.org/10.1007/BFb0069184}
{\it The Spectrum of a Graph}. 
In: A. Dold, B. Eckmann, C.H.C Little (eds.), Lect. Notes. Math.,
{\bf 622}, Springer, Berlin, Heidelberg, 1977. MR0544356

\bibitem{ImriKlav} W. Imrich, S. Klavzar, 
\href{https://www.amazon.com/Product-Graphs-Recognition-Wilfried-Imrich/dp/0471370398}
{\it Product graphs}, Wiley-Interscience Series in Discrete Mathematics and Optimization. Wiley-Interscience, New York, 2000.

\bibitem{Isang2000} R.R. Isangulov, {\it Isospectral flat Klein bottles} (Russian), Mat. Zamet. YAGU, {\bf 7}:2 (2000), 39--48. Zbl 0983.58016

\bibitem{Kac} M. Kac, 
\href{https://doi.org/10.2307/2313748}
{\it Can one hear the shape of a drum?}, 
Amer. Math. Monthly 73 (1966), no. 4, 1--23. MR201237

\bibitem{Kitaoka} Y. Kitaoka, \href{https://doi.org/10.1007/BF01223956} {\it Positive definite quadratic forms with the same representation numbers}, Arch. Math. (Basel), {\bf 28}:5 (1977), 495--497. MR441864

\bibitem{Kneser} M. Kneser, \href{https://doi.org/10.1007/BF01361543}{\it Lineare Relationen zwischen Darstellungsanzahlen quadratischer Formen} (German), Math. Ann., {\bf 168} (1967), 31--39. MR205943
  
\bibitem{LinWanZhang} Y. Lin, S. Wan, H. Zhang,
\href{https://arxiv.org/abs/2403.06105}
{\it Connection Laplacian on discrete tori with converging property},
arXiv preprint arXiv:2403.06105, 2024 - arxiv.org

\bibitem{LiuLu} Xiaogang Liu,  Pengli Lu, 
\href{https://doi.org/10.1142/S1793830916500282}
{\it Laplacian spectral characterization of dumbbell graphs and theta graphs}, 
Discrete Math. Algorithms Appl., {\bf 8}:2 (2016) 1650028 (10 pages)  MR3505475

\bibitem{Louis} J. Louis, 
\href{https://doi.org/10.1016/j.ejc.2017.03.003}
{\it Asymptotics for the determinant of the combinatorial Laplacian on hypercubic lattices}, 
European J. Comb., {\bf 63} (2017), 176--196. MR3645793

\bibitem{MedMedTheta} A. Mednykh, I. Mednykh, 
\href{https://doi.org/10.26493/1855-3974.550.e1a}
{\it Isospectral genus two graphs are isomorphic},
Ars Math. Contemp. {\bf 10}:2 (2015), 223--235. MR3529288

\bibitem{Milnor64} J. Milnor,  
\href{https://doi.org/10.1073/pnas.51.4.542}
{\it Eigenvalues of the Laplace operator on certain manifolds}, 
Proc. Nat. Acad. Sci. U.S.A. {\bf 51} (1964), 542. MR162204

\bibitem{Mohar91} B. Mohar, 
\href{https://www.amazon.com/Graph-Theory-Combinatorics-Applications-Yousef/dp/0471532452} 
{\it The Laplacian spectrum of graphs}, in Graph theory, combinatorics, and applications {\bf 2}, Ed.  Y. Alavi, G. Chartrand, O.R. Oellermann, A.J. Schwenk, Wiley, New York (1991), 871--898. MR1170831

\bibitem{NilRowRyd} E. Nilsson,  J. Rowlett,  F. Rydell, 
\href{https://doi.org/10.1090/bull/1770} 
{\it The isospectral problem for flat tori from three perspectives}, 
Bull.  Amer.  Math.  Soc. (New Series), {\bf 60}:1 (2023), 39--83. MR4520776

\bibitem{Sabidu} G. Sabidussi, 
\href{https://doi.org/10.1007/BF01162967}
{\it Graph multiplication}, 
Math. Z., {\bf 72}  (1960), 446--457. (1963). MR0209177

\bibitem{Schiemann} A. Schiemann, \href{https://doi.org/10.1007/BF01189584}
{\it Ein Beispiel positiv definiter quadratischer Formen der Dimension $4$ mit gleichen Darstellungszahlen} (German), Arch. Math. (Basel), {\bf 54}:4 (1990), 372--375. MR1042130

\bibitem{Shiota}  K.-i. Shiota, 
\href{https://doi.org/10.1215/kjm/1250519669}
{\it On theta series and the splitting of $S_2(\Gamma_0(q))$}, 
J. Math. Kyoto Univ., {\bf 31}:4 (1991), 909--930. MR1141077
 
\bibitem{Vizing} 
V. G. Vizing, 
\href{https://www.semanticscholar.org/paper/The-cartesian-product-of-graphs-Vizing/1af5fa6fd4cdb43baf9203d85015cae5eef2e5ea}
{\it The Cartesian product of graphs} (Russian). Vychisl. Sistemy, {\bf 9}, (1963), 30--43. MR0209178
English translation in Comp. El. Syst. {\bf 2} (1966), 352--365.

\bibitem{Wolpert} S. Wolpert, 
\href{https://doi.org/10.2307/1971114}
{\it The length spectra as moduli for compact Riemann surfaces}, 
Ann. Math. (2), {\bf 109}:2 (1979), 323--351. MR0528966

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}




\thispagestyle{empty}

\title[One can hear a discrete rectangular torus]{\large One can hear a discrete rectangular torus}

\author[A.D. MEDNYKH]{{\bf A.D. MEDNYKH}\protect\orcidicon{0000-0001-8827-2630}}
\author[I.A. MEDNYKH]{{\bf I.A. MEDNYKH}\protect\orcidicon{0000-0001-8827-2630}}
\author[G.K. SOKOLOVA]{{\bf G.K. SOKOLOVA}\protect\orcidicon{0000-0001-8827-2630}}


\address{Alexander Dmitrievich Mednykh
% Êîíòàêòíûå äàííûå âñåõ àâòîðîâ è ìåñòî ðàáîòû óêàçûâàþòñÿ òîëüêî íà àíãëèéñêîì ÿçûêå
\newline\hphantom{iii} Sobolev Institute of Mathematics,
\newline\hphantom{iii} pr. Koptyuga, 4,
\newline\hphantom{iii} 630090, Novosibirsk, Russia}%
\email{\textcolor{blue}{mednykh@math.nsc.ru}}%

\address{Ilya Aleksandrovich Mednykh
% Êîíòàêòíûå äàííûå âñåõ àâòîðîâ è ìåñòî ðàáîòû óêàçûâàþòñÿ òîëüêî íà àíãëèéñêîì ÿçûêå
\newline\hphantom{iii} Sobolev Institute of Mathematics,
\newline\hphantom{iii} pr. Koptyuga, 4,
\newline\hphantom{iii} 630090, Novosibirsk, Russia}%
\email{\textcolor{blue}{i.a.mednykh@math.nsc.ru}}%

\address{Galina Konstantinovna Sokolova
% Êîíòàêòíûå äàííûå âñåõ àâòîðîâ è ìåñòî ðàáîòû óêàçûâàþòñÿ òîëüêî íà àíãëèéñêîì ÿçûêå
\newline\hphantom{iii} Sobolev Institute of Mathematics,
\newline\hphantom{iii} pr. Koptyuga, 4,
\newline\hphantom{iii} 630090, Novosibirsk, Russia}%
\email{\textcolor{blue}{98gal@mail.ru}}%

\thanks{\sc Mednykh, A.D., Mednykh I.A., Sokolova G.K., 
One can hear a discrete rectangular torus}
\thanks{\copyright \ 2023 Mednykh, A.D., Mednykh I.A., Sokolova G.K.}
\thanks{\rm The work is done in the framework by the State Contract of the Sobolev Institute of Mathematics (project no. FWNF--2022--0005)}   
\thanks{\it Received January, 1, 2023, Published December, 31, 2023}%


\semrtop \vspace{1cm}
\maketitle {\small

\vspace{-12pt}
\centerline{{\it Communicated by} {\sc  A.V. Pyatkin}
%({\it filled by Editorial board})
}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:} In the present paper, we prove that two discrete rectangular tori are isospectral if and only if they are isomorphic.
\medskip

\noindent{\bf Keywords:} Laplacian matrix, isospectral manifolds, isospectral graphs, discrete torus.
\end{quote}}
\bigskip

\section{Introduction}
 
In 1964, in his one-page paper ``Eigenvalues of the Laplace Operator on Some Manifolds'', J. Milnor \cite{Milnor64} described an example of two 16-dimensional flat tori that are isospectral but not isometric. Milnor's article inspired Mark Kac to write his famous paper \cite{Kac} entitled ``Can one hear the shape of a drum?''. Since these classical works, the question of what geometric properties of a manifold are determined by its Laplace operator has been the source of much research. Wolpert \cite{Wolpert} showed that a generic Riemann surface is determined by its Laplace spectrum. Nevertheless, pairs of isospectral non-isometric Riemann surfaces in every genus $\ge 4$ are known. See papers by Buser \cite{Buser86}, Brooks and Tse \cite{BrooksTse87}, and others. There are also examples of isospectral non-isometric surfaces of genus two and three with variable curvature Barden and Hyunsuk Kang \cite{BardKang}. At the same time,  R. Brooks \cite{Brooks88} proved that isospectral genus one Riemann surfaces (flat tori) are isometric. A similar result for the Klein bottle was obtained by R. Isangulov \cite{Isang2000}.

As we have seen, one-dimensional isospectral flat tori are always isometric, so a natural question is, What is the lowest dimension in which there are isospectral non-isometric flat tori? Following Milnor, the search for pairs of isospectral but non-isometric flat tori became a race towards the lowest possible dimension. Kneser found a 12-dimensional example in 1967 \cite{Kneser}. Ten years later, Kitaoka \cite{Kitaoka} reduced this to eight. In 1986, Conway and Sloane \cite{ConwaySloan} found five- and six-dimensional examples. In 1990, Schiemann \cite{Schiemann} constructed a four-dimensional example. Independently, and using a different approach, Shiota \cite{Shiota} found another example one year later in 1991. The same year, Earnest and Nipp \cite{EarnNipp} contributed one more pair.

Similar results are also known for graphs.  The problem of how graphs are defined by their spectrum is 
the subject  of many  studies (see survey by E.R.van Dam and W.H.Haemers \cite{DamHaemers}).
Peter Buser \cite{Buser92} posed an interesting problem: are two isospectral Riemann surfaces of genus two isometric? Up to our knowledge the problem is still open but, quite probably, can be solved positively. Because of the intrinsic link between Riemann surfaces and graphs, we hope that our result will be helpful to make a progress in solution of the Buser problem. More precisely, in the papers \cite{MedMedTheta} and \cite{LiuLu} the discrete counterpart for genus two Riemann surface was considered. This role is played by the so-called theta graphs. In these papers, In these papers, it was shown that two isospectral theta graphs are isomorphic.   

Recently, the following result was obtained. Consider a \textit{rectangular lattice} $\Gamma$ in $\mathbb{R}^{n}$ with diagonal basis matrix. Define a \textit{rectangular flat torus} as the associated flat torus $\mathbb{R}^{n}/\Gamma.$ Then, by (\cite{NilRowRyd}, Theorem 2.35), two rectangular flat tori are isospectral if and only if they are isometric. 

The aim of the present paper is to establish the similar result for discrete rectangular tori.

\section{Preliminary results and definitions}

\subsection{Laplacian for graphs and isospectrality} 

The graphs in this paper are undirected. Denote by $V(G)$ and $E(G),$ respectively, the set of vertices and edges of a graph $G.$ For each $u,v \in V(G),$ we set $a_{u\,v}$ to be equal to the number of edges between vertices $u$ and $v.$ The matrix $A=A(G)= \{a_{u\,v}\}_{u,v\in V(G)}$ is called the \textit{adjacency matrix} of the graph $G.$ Let $d(v)$ denote the valency of $v\in V(G)$,  and let $D=D(G)$ be the diagonal matrix indexed by $V(G)$ and with $d_{v\,v}=d(v).$

The matrix $L=L(G)=D(G)-A(G)$ is called the {\it Laplacian matrix} of $G.$ The Laplacian matrix $L(G)$ is a $n\times n$ square matrix, where $n = |V(G)|$ is the number of vertices in $G.$ Throughout the paper we shall denote by $\chi(G,x)$ the characteristic polynomial of $L(G).$ That is $\chi(G,\lambda)=\det(L(G)-\lambda I_{n}),$ where $I_{n}$ is the identity $n\times n$ matrix. For brevity, we will call $\chi(G,x)$ the \textit{Laplacian polynomial} of $G.$ The roots of $\chi(G,x)$ will be called the \textit{Laplacian eigenvalues} (or just eigenvalues) of $G.$ They are all non-negative numbers and can be arranged as follows 
$$0=\lambda_{1}(G)\le\lambda_{2}(G)\le\ldots\le\lambda_{n}(G),$$ repeated according to their multiplicity. Recall that for connected graph $G$ we always have $\lambda_{1}(G)=0$ and $\lambda_{2}(G) > 0.$

Two graphs $G$ and $H$ are called \textit{isospectral} if their Laplacian polynomials coincide $\chi(G,\lambda) = \chi(H,\lambda).$

\subsection{Cartesian product and uniqueness of decomposition}

The Cartesian product of graphs $G$ and $H$ is a graph, denoted $G\times H,$ whose vertex set is $V (G) \times V (H).$ Two vertices $(g, h)$ and $(g', h')$ are adjacent if either $g = g'$ and $h$ is adjacent to $h'$ in $H,$ or $h = h'$ and $g$ is adjacent to $g' $ in $G.$  A non-trivial graph $G$ is called {\it prime} if $G = G_1\times G_2$ implies that either $G_1$ or $G_2$ is a one vertex graph $K_1.$ It is easy to see that every non-trivial finite graph has a prime factorization with respect to the Cartesian product. The number of prime factors is at most $\log_2 |V (G)|.$

The uniqueness of the prime factor decomposition of connected graphs with respect to the Cartesian product was first shown by Sabidussi \cite{Sabidu}, and independently by Vizing \cite{Vizing}. More precisely, the following theorem holds.

\begin{theorem}[\textbf{Sabidussi-Vizing}]
A connected graph has a unique representation as a product of prime graphs, up to isomorphism and the order of the factors.
\end{theorem}

See book (\cite{ImriKlav}, Charter 6) for the proof of this theorem. 

\subsection{Theta function of graphs}

Let $G$ be a finite graph. Denote by $S_{G}$ the Laplacian spectrum of $G.$ Define $\Theta_{G}(t)$ function of graph $G$ in the following way
$$\Theta_{G}(t)= \sum\limits_{\lambda\in S_{G}}e^{-\lambda t}.$$ 

According to R. Brooks \cite{Brooks88}, we note that function $\Theta_{G}(t)$ completely defines spectrum of $G.$ Indeed, suppose that 
$$\Theta_{G}(t)=c_{1}\,e^{-\mu_{1}t}+c_{2}\,e^{-\mu_{2} t}+\ldots+c_{\ell}\,e^{-\mu_{\ell} t},\,0 \le \mu_{1}<\mu_{2} < \ldots < \mu_{\ell}.$$ Here $\mu_{j}$ are all elements of $S_{G}$ taken without repetitions and $c_{j}$ theirs respective multiplicities.

The multiplicities $c_{j}$ of eigenvalues $\mu_{j}$ are completely defined by $\Theta_{G}(t),$ because all functions $e^{-\mu_{j} t}$ are linearly independent. To find exact values of $c_{j}$ and $\mu_{j}$ we consider the expression 
$$e^{\mu t}\Theta_{G}(t)=c_{1}e^{(\mu - \mu_{1}) t} +c_{2}e^{(\mu - \mu_{2}) t} +\ldots+ 
  c_{\ell}e^{(\mu - \mu_{\ell}) t}.$$

We can conclude that $\mu_{1}=\underset{\mu}{\max}\{\underset{t\to\infty}{\lim}(e^{\mu t}\Theta_{G}(t))\}\text{ is finite}.$  Putting $\mu=\mu_{1}$ we have $\underset{t\to\infty}{\lim}e^{\mu t}\Theta_{G}(t)=c_{1}.$ 

Assume that we already found the initial values $(c_{1},\mu_{1}), (c_{2}, \mu_{2}), \ldots, (c_{k}, \mu_{k})$ for $k<\ell.$ Then $\mu_{k+1}$ is equal to $\underset{\mu}{\max}\{\underset{t\to\infty}{\lim}(e^{\mu t}(\Theta_{G}(t)-\sum\limits_{j=1}^{k}c_{j}e^{-\mu_{j} t}))\}\text{ is finite}.$ Its corresponding multiplicity $c_{k+1}$ is equal to $\underset{t\to\infty}{\lim}e^{\mu_{k+1}t}(\Theta_{G}(t)-\sum\limits_{j=1}^{k}c_{j}e^{-\mu_{j} t}).$

As a result, we get the following lemma.

\begin{lemma}\label{isolemma}
Two finite graphs $G_{1}$ and $G_{2}$ are isospectral if and only if $\Theta_{G_{1}}(t)=\Theta_{G_{2}}(t).$
\end{lemma}

Moreover, we can derive one more important property of $\Theta$-function for graphs. 

\begin{lemma}\label{thetalemma}
Let $G_1$ and $G_2$ are two finite graphs. Denote by $G_{1}\times G_{2}$ the Cartesian product of $G_{1}$ and $G_{2}.$ Then $\Theta_{G_{1}\times G_{2}}(t)=\Theta_{G_{1}}(t)\,\Theta_{G_{2}}(t).$ 
\end{lemma}

\begin{proof}
Let $\lambda_{1},\lambda_{2},\ldots,\lambda_{m}$ be the spectrum of $G_{1}$ and $\mu_{1},\mu_{2},\ldots,\mu_{n}$ be the spectrum of $G_{2}$ By \cite{Mohar91} the spectrum of Cartesian product $G_{1}\times G_{2}$ is given by the set $\{\lambda_{i}+\mu_{j},\,1\le i\le m,1\le j\le n\}.$ 

Let $G=G_{1}\times G_{2}.$ We have

$$\Theta_{G}(t)= \sum\limits_{\lambda\in S_{G}}e^{-\lambda t}= \sum\limits_{i=1}^m \sum\limits_{j=1}^n e^{-(\lambda_{i}+\mu_{j}) t}=\sum\limits_{i=1}^m e^{-\lambda_{i} t}\sum\limits_{j=1}^n  e^{-\mu_{j}t}=\Theta_{G_1}(t)\,\Theta_{G_2}(t).$$ 
\end{proof}

\subsection{Discrete tori} We define a {\it discrete rectangular torus} as the Cartesian product $C_{m_1}\times C_{m_2}\times\ldots \times C_{m_p},$ where $C_m$ is a cyclic graph on $m$ vertices. Excluding trivial factors we assume that $m_i\ge  2,\, i=1,2,\ldots,p.$ In what follows, instead of the long term ``discrete rectangular  torus'', we will freely use the terms ``discrete torus'' or simply ``torus''.

The Cartesian product of graphs is commutative and associative in the following sense. For any two graphs 
$G_1,G_2$ the products  $G_1\times G_2$ and $G_2\times G_1$ are isomorphic. Given graphs $G_1,G_2$ and $G_3,$ the map $ ((x_1,x_2),x_3) \to (x_1,(x_2,x_3))$ is an isomorphism $(G_1 \times G_2) \times G_3 \to G_1 \times (G_2 \times G_3).$

Associativity and commutativity gives us permission to omit parentheses when dealing with products with more than two factors.This allows to present a discrete rectangular torus in the following  form
$$T=T_{m_1,\ldots,m_{p}}=C_{m_1}\times C_{m_2}\times\ldots \times C_{m_p},   \text{  where  }\,\,
1<m_1\le m_2\le\ldots\le m_p.$$

By making use of Sabidussi-Vizing  theorem, one can deduce that the latter representation of $T$ is unique.  Indeed, by Unique Square Lemma (\cite{ImriKlav}, Lemma 6.3), every nontrivial Cartesian product $G_1\times G_2$ contains a square subgraph $C_4=K_2\times K_2$. Thus, a cyclic graph $C_m$ is prime for any $m\neq4.$ 
Then, by replacing $C_4$ with two copies of $K_2$ in the above representation, we obtained an irreducible Cartesian product of prime graphs. According to the Sabidussi-Vizing theorem, it is unique. Replacing back the double product $K_2\times K_2$ with one copy of $C_4$ we have the uniqueness of the required expression.

For convenience, we will call the parameter $p$ the {\it dimension} of the  torus $T.$ As a result, we have the following important property.

\begin{lemma} Any discrete rectangular torus $T$  of dimension $p$ can be uniquely represented in the form 
$$T=C_{m_1}\times C_{m_2}\times\ldots\times C_{m_p},\,\text{ where }\,1<m_1\le m_2\le\ldots\le m_p.$$
\end{lemma}

\subsection{Spectrum of discrete torus}
We will use the following observation from \cite{Mohar91}. The Laplacian eigenvalues of the Cartesian product $G_1\times G_2$ of graphs $G_1$ and $G_2$ are equal to all the possible sums of eigenvalues of the two factors: $\lambda_{i}(G_{1})+\lambda_{j}(G_{2}),\,i = 1,\ldots , |V (G_1 )|,\,j = 1, \ldots, |V (G_2 )|.$

Let $G$ be a connected graph on $n$ vertices and 
$$0=\lambda_1 (G )<\lambda_2 (G )=a(G)\le\lambda_3 (G )\le\ldots\le\lambda_n (G )$$
be its Laplacian spectrum. Let us call the second smallest eigenvalue $a(G)$ of the spectrum an {\it algebraic connectivity} of the graph $G.$ Following to \cite{Fiedler}, from the above observation we note that $a(G_{1}\times G_{2})=\min(a(G_1),\,a(G_2)).$ 

We set $G_i=C_{m_i},$ for $i=1,\ldots,\,p.$ Then for any $i$ spectrum of $G_i$ is given by the list
$$\{ 4\sin^2(\frac{\pi j}{m_i}),\,j=0,1,\ldots, m_i-1\}.$$  
Hence, $a(G_i)=4\sin^2(\frac{\pi }{m_i}).$ Now, consider the discrete torus
$$T=T_{m_1,\ldots,m_{p}}=C_{m_1}\times C_{m_2}\times\ldots \times C_{m_p},\text{ where }
1<m_1\le m_2\le\ldots\le m_p.$$
Then, because of associativity of Cartesian product, spectrum of $T$ is given by
$$\{\sum\limits_{i=1}^p 4\sin^2(\frac{\pi j_i}{m_i}),\,i=1,\ldots,p,\,j_i=0,1,\ldots, m_i-1\}.$$ 
Also,
$$a(T)=\min(a(G_1), a(G_2),\ldots, a(G_p))=4\sin^2(\frac{\pi }{m_p}).$$

\subsection{One can hear the dimension of a discrete torus} 
In this point, we prove that one can hear the dimension of a discrete torus. In other words, if two tori are isospectral, then they have the same dimension. More precisely, we get the following statement. 

\begin{proposition}\label{dimprop} 
Consider two isospectral disctere rectangular tori $T_{m_1,\ldots,m_p},\linebreak 1<m_1\le\ldots\le m_p$ and $T_{\tilde{m}_1,\ldots,\tilde{m}_{\tilde{p}}},\,1< \tilde{m}_1\le \ldots\le \tilde{m}_{\tilde{p}}.$ Then $p=\tilde{p}.$
\end{proposition} 

\begin{proof} If $p=\tilde{p},$ we have what needs to be done. Without loss of generality, we can assume that $p<\tilde{p}.$ Since the tori  $T=T_{m_1,\ldots,m_p}$ and $\tilde{T}= T_{\tilde{m}_1,\ldots,\tilde{m}_{\tilde{p}}}$
are isospectral, they share the algebraic connectivity $a(T)=a(\tilde{T}),$ which is certainly a spectral invariant. Therefore, $4\sin^2(\frac{\pi }{m_p})=4\sin^2(\frac{\pi }{\tilde{m}_{\tilde{p}}}),$ that gives us $m_p=\tilde{m}_{\tilde{p}}.$

Recall the number of vertices $m_1\cdot\ldots\cdot m_p$ of the torus $T_{m_1,\ldots,m_p}$
is also a spectral invariant.

We represent tori $T$ and $\tilde{T}$ as the Cartesian product $T=T_{m_1,\ldots,m_{p-1}}\times C_{m_p}$
and $\tilde{T}=T_{\tilde{m}_1,\ldots,\tilde{m}_{\tilde{p}-1}}\times C_{m_p}.$ Since $T$ and $\tilde{T}$ are isospectral, by Lemma~\ref{isolemma} we have $\Theta_{T}(t)=\Theta_{\tilde{T}}(t).$ By Lemma~\ref{thetalemma}, $\Theta_{T}(t)=\Theta_{A}(t)\Theta_{C_{m_p}}(t)$ and $\Theta_{\tilde{T}}(t)=\Theta_{B}(t)\Theta_{C_{m_p}}(t),$ where $A=T_{m_1,\ldots,m_{p-1}}$ and $B=T_{\tilde{m}_1,\ldots,\tilde{m}_{\tilde{p}-1}}.$ Since $\Theta_{C_{m_p}}(t)$ is strictly positive, we have $\Theta_{A}(t)=\Theta_{B}(t).$  That is, graph $A$ and $B$ are isospectral and $a(A)=a(B).$ Hence, $m_{p-1}=\tilde{m}_{\tilde{p}-1}$ and, by the same arguments, $T_{m_1,\ldots,m_{p-2}}$ and $ T_{\tilde{m}_1,\ldots,\tilde{m}_{\tilde{p}-2}}$ are isospectral. Repeating the process, we obtain that $m_{1}=\tilde{m}_{\tilde{p}-p+1}$ and tori $T_{m_1}$ and $ T_{\tilde{m}_1,\ldots,\tilde{m}_{\tilde{p}-p+1}}$ are isospectral. The latter is impossible, since the number of vertices $m_1$ of the first torus is strictly less than $\tilde{m}_1\cdot\ldots\cdot\tilde{m}_{\tilde{p}-p+1}$ of the second one.
\end{proof}


\section{Main results}

The main result of the paper is the following theorem.
\begin{theorem} \label{mainth}
Two discrete rectangular tori are isospectral if and only if they are isomorphic.     
\end{theorem}
\begin{proof} If two tori are isomorphic, then up to the enumeration of vertices they have the same Laplace matrix and, therefore, have the same Laplace spectrum. That is, they are isospectral.

Conversely, suppose that two discrete tori $T_{m_1,\ldots,m_p},\,1< m_1 \le\ldots\le m_p$ and 
$T_{\tilde{m}_1,\ldots,\tilde{m}_{\tilde{p}}},\,1< \tilde{m}_1\le \ldots\le \tilde{m}_{\tilde{p}}$ are isospectral. From Proposition~\ref{dimprop} we already know that $p=\tilde{p}.$ 

Now, to proof the isomorphism between $T=T_{m_1,\ldots,m_p}$ and $\tilde{T}=T_{\tilde{m}_1,\ldots,\tilde{m}_p}$ we can use induction on the dimension $p.$ For $p=1,$ isospectral tori $T_{m_1}=C_{m_1}$ and $T_{\tilde{m}_1}=C_{\tilde{m}_1}$ are cyclic graphs with the same number of vertices $m_1=\tilde{m}_1$ and, therefore, are isomorphic.  

By induction, suppose that the theorem is proved for all $k<p,$ where $p\ge 2.$ Let us prove it for $p=k.$ We  represent tori $T$ and $\tilde{T}$ as the Cartesian product  $T=A\times C_{m_p}$ and $\tilde{T}=B\times C_{\tilde{m}_p},$ where $A=T_{m_1,\ldots,m_{p-1}}$ and $B=T_{\tilde{m}_1,\ldots,\tilde{m}_{p-1}}$ are respective $(p-1)$-dimensional tori. Repeating the arguments in the proof of Proposition~\ref{dimprop}, we note that the minimal non-zero eigenvalue is common to  graphs $T$ and $\tilde{T}$ and is equal to $4\sin^2(\frac{\pi }{m_p})=4\sin^2(\frac{\pi }{\tilde{m}_p}).$ Hence, $m_p=\tilde{m}_p.$ Since $T$ and $\tilde{T}$ are isospectral, by Lemma~\ref{isolemma} we have $\Theta_{T}(t)=\Theta_{\tilde{T}}(t).$ By Lemma~\ref{thetalemma}, $\Theta_{T}(t)=\Theta_{A}(t)\Theta_{C_{m_p}}(t)$ and $\Theta_{\tilde{T}}(t)=\Theta_{B}(t)\Theta_{C_{m_p}}(t).$ Since $\Theta_{C_{m_p}}(t)>0,$ we have $\Theta_{A}(t)=\Theta_{B}(t).$  That is, graph $A$ and $B$ are isospectral. By inductive hypothesis, they are isomorphic. Then $T=A\times C_{m_p}$ and $\tilde{T}=B\times C_{m_p}$ are also isomorphic. 
\end{proof}

\section{Conclusion}

The simplicity of the proof of Theorem~\ref{mainth} is deceptive. Such statements do not hold true even for graphs with a simple cyclic structure. For example, we can consider circulant graph $G=C_{n}(s_1,\,s_2,\cdots,s_k),$ where  $0<s_1<s_2<\ldots<s_k<n/2.$ Its Laplacian spectrum is well known. It is given by the list $S_G=\{\sum_{i=1} ^k 4\sin^2(\frac{\pi s_i j}{n}),\,j=0,1,\ldots,n-1\}.$ It was shown  by Godsil, Holton and McKay \cite{Godsil}, that graphs $C_1=C_{20}(2,3,4,7)$ and $C_2=C_{20}(3,6,7,8)$ are isospectral but not isomorphic.  
Moreover, there are no cospectral circulant graphs on less than twenty vertices.

The discrete rectangular torus considered in this paper is the simplest discrete analogue of a continuous multidimensional torus. There is a more general aspect of the discrete torus that emulates the general continuous torus coming from a multidimensional parallelepiped. See, for instance papers \cite{Louis} and \cite{LinWanZhang}. The structure of Laplacian operator as well as its spectrum is known in this case. Up to our knowledge, the problem of isospectrality of such tori has not been solved.
 
The same problem naturally arises for connection Laplacian in vector bundles on discrete tori. The properties of the connection Laplacian are the subject of investigation of many recent papers \cite{Friedli}, \cite{LinWanZhang} and others.

\bigskip

\end{document}
